Author's response to reviews

Title: Clinical efficacy of B-mode ultrasound-guided percutaneous transhepatic gallbladder drainage followed by laparoscopic cholecystectomy for acute cholecystitis in high-risk elderly patients

Authors:

Yi-Ren Hu (huyiren2014@163.com)
Jiang-Hua Pan (panjianghua2014@163.com)
Xiao-Chun Tong (tongxiaochun2014@163.com)
Ke-Qin Li (likeqin2014@163.com)
Sen-Rui Chen (chensenrui2014@126.com)
Yi Huang (huangyi1029@126.com)

Version: 2 Date: 23 December 2014

Author's response to reviews:

The line and page numbers are added. Page breaks are removed.